Author's response to reviews

Title: Cytotoxicity Screening of Bangladeshi Medicinal Plant Extracts on Pancreatic Cancer Cells

Authors:

Sherine George (sgeorge4@illinois.edu)
Siddharth V Bhalerao (sbhaler2@illinois.edu)
Erich A Lidstone (elidst2@illinois.edu)
Irfan S Ahmad (isahmad@illinois.edu)
Atiya Abbasi (atiya.abbasi@iccs.edu)
Brian T Cunningham (bcunning@illinois.edu)
Kenneth L Watkin (watkin@illinois.edu)

Version: 5 Date: 30 August 2010

Author's response to reviews: see over
August 30th, 2010
Biomed Central Editorial Board

Dear Editors,

**SUB: Manuscript revisions**

Please consider our revised manuscript (no. 1348472814366240) titled “Cytotoxicity Screening of Bangladeshi Medicinal Plant Extracts on Pancreatic Cancer Cells”.

We have made the recommended change in the manuscript in response to the second reviewer’s comments as discussed below. We have also made all the recommended format changes to the manuscript. We thank the reviewers and the editors once again for their helpful input and hope that these revisions sufficiently address all comments.

**Reviewer 2 (Ramzi Mothana):**

1- Full names of the plants (complete species names with authority) should be added in table 1. It is also better to write the voucher numbers of the plants.

**Response:** We have updated Table 1 to include complete species name with authority. We still do not presently have a complete list of voucher numbers and are unable to include this information.

**Recommended format changes:**

1. Title - Please use sentence case in the title, both in the manuscript file and the submission system (i.e. remove all unnecessary capitals).

2. Conflicts of interest statement - Please rename this section Competing interests.

3. Author contributions - Please rename this section Authors' contributions.

4. Acknowledgements - Please place this section after the Authors' contributions.

**Response:** We have now made all the format changes listed above.

Thanking you,
Sincerely
Kenneth L. Watkin
(Corresponding Author)